Author's response to reviews

Title: Hypomania spectrum disorder in adolescence: A 15-year follow-up of non-mood morbidity in adulthood

Authors:

Aivar Päären (aivar.paaren@neuro.uu.se)
Hannes Bohman (hannesbohman@yahoo.se)
Anne-Liis von Knorring (anne-liis.von_knorring@neuro.uu.se)
Lars von Knorring (lars.von_knorring@neuro.uu.se)
Gunilla Olsson (ig.olsson@neuro.uu.se)
Ulf Jonsson (ulf.jonsson@neuro.uu.se)

Version: 4 Date: 3 January 2014

Author's response to reviews:

Uppsala, 03 JAN 2014

Ms. Sarah Doucette
Editor, BioMed Central Psychiatry

Dear Editor Doucette

Many thanks for the review of our revised manuscript MS: 4126592501109290 “Hypomania spectrum disorder in adolescence: A 15-year follow-up of non-mood morbidity in adulthood”. We are delighted that our revision is satisfactory.

The manuscript has now been edited for language and grammar by the Edanz Group Ltd. and we believe that it is now suitable for publication in the BMC Psychiatry.

Sincerely yours,

Aivar Päären